Author’s response to reviews

Title: Advancing pandemic influenza control in the general population: qualitative research perspectives from Australia and the UK

Authors:

Mark D M Davis Dr (mark.davis@monash.edu)
Niamh Stephenson Dr (n.stephenson@unsw.edu.au)
Davina Lohm Dr (davina.lohm@monash.edu)
Emily Waller Ms (e.waller@unsw.edu.au)
Paul Flowers Prof (p.flowers@gcu.ac.au)

Version: 2
Date: 30 September 2014

Author’s response to reviews:

Dear Editors

As requested, the manuscript is now line numbered and the methods section makes note of written informed consent.

Yours sincerely

Mark Davis